Author’s response to reviews

Title: The effect of methamphetamine abuse on dental caries and periodontal diseases in an Eastern China city

Authors:

Tao Ye (461381831@qq.com)
Dongliang Sun (sdlfmmu@163.com)
Guangying Dong (donggy@fmmu.edu.cn)
Guangjie Xu (guangjiexu@126.com)
Ligang Wang (308594619@qq.com)
Jinjin Du (308594618@qq.com)
Pengcheng Ren (tdrpch@fmmu.edu.cn)
Shibin Yu (yushibin@fmmu.edu.cn)

Version: 4 Date: 07 Sep 2017

Author’s response to reviews:

BMC Oral Health

MS: OHEA-D-17-00085R3

Title: The effect of methamphetamine abuse on dental caries and periodontal diseases in an Eastern China city

Tao Ye, Dongliang Sun, Guangying Dong, Guangjie Xu, Ligang Wang, Jinjin Du, Pengcheng Ren *, Shibin Yu *

Response to Editor’s Comments

[Point 1]: Please can the specific p values be reported in the abstract not p<0.05

[Response]: Thank you for your suggestion! The specific p values has been given in Abstract according to your suggestion. Please see the revised manuscript Page 4 Line 2, 4.

[Point 2]: Suggested title for Table 1: Association between Caries, periodontal status and age, drug abuse duration and tooth brushing habits of former meth users (N=162)
[Response]: Good idea! The title for Table 1 has been revised according to your suggestion. Please see the revised manuscript Page 23 Line 1.

[Point 3]: The first variable on tooth brushing should be tooth brushing frequency and not tooth brushing time

[Response]: The word “Tooth brushing times” has been replaced with “Tooth brushing frequency” in Table 1. Please see the revised manuscript Page 23 Line 8.